Author’s response to reviews

Title: SESOTHO trial ("Switch Either near Suppression Or THOUSand") - Switch to second-line versus WHO-guided standard of care for unsuppressed patients on first-line ART with viremia below 1000c/mL: protocol of a multicenter, parallel-group, open-label, randomized controlled trial in Lesotho, Southern Africa

Authors:

Alain Amstutz (alain.amstutz@unibas.ch)
Bienvenu Nsakala (b.nsakala@solidarmed.ch)
Fiona Vanobberghen (fiona.vanobberghen@swisstph.ch)
Josephine Muhairwe (j.muhairwe@solidarmed.ch)
Tracy Glass (tracy.glass@swisstph.ch)
Beatrice Achieng (b.achieng@solidarmed.ch)
Mamorena Sepeka (ssepeka@gmail.com)
Katleho Tlali (k.tlali@solidarmed.ch)
Lebohang Sao (lebohangchere@gmail.com)
Kyaw Thin (dr.kyawthin@gmail.com)
Thomas Klimkait (thomas.klimkait@unibas.ch)
Manuel Battegay (manuel.battegay@usb.ch)
Niklaus Labhardt (n.labhardt@unibas.ch)

Version: 3 Date: 17 Jan 2018

Author’s response to reviews:

To Editor-in-Chief
Dr Cecilia Devoto
BMC Infectious Diseases
BMA House, Tavistock Square,
Submission of the following second minor revision of the study protocol: SESOTHO trial („Switch Either near Suppression Or THOusand”) – Switch to second-line versus WHO-guided standard of care for unsuppressed patients on first-line ART with viremia below 1000 c/mL: protocol of a multicenter, parallel-group, open-label, rando-mized controlled trial in Lesotho, Southern Africa

Dear Dr Cecilia Devoto

We thank you for reviewing our manuscript and for considering a revised version for publication in BMC Infectious Diseases. We have addressed the formatting points raised by you.

We hope that the revised manuscript meets your expectations and that it can be accepted for publication in the present form.

Thank you very much.

Sincerely yours,

Alain Amstutz, MSc
cMD-PhD Clinical Research,
Swiss TPH (Clinical Research Unit)
& University Hospital Basel (Division of Infectious Diseases)
University of Basel
Socinstrasse 57, 4051 Basel, Switzerland

Dr Niklaus Labhardt, MD MIH
Principal Investigator,
Head of „Towards 90-90-90 in Lesotho“
research consortium,
Swiss TPH
Socinstrasse 57, 4051 Basel, Switzerland